

Framework for Network-Constrained Tracking of Cyclists and Pedestrians

Vial, Alphonse; Hendeby, Gustaf; Daamen, Winnie; van Arem, Bart; Hoogendoorn, Serge

DOI 10.1109/TITS.2022.3225467

Publication date 2022 **Document Version** Final published version

Published in IEEE Transactions on Intelligent Transportation Systems

Citation (APA) Vial, A., Hendeby, G., Daamen, W., van Arem, B., & Hoogendoorn, S. (2022). Framework for Network-Constrained Tracking of Cyclists and Pedestrians. *IEEE Transactions on Intelligent Transportation Systems*, 24(3), 3282-3296. https://doi.org/10.1109/TITS.2022.3225467

Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright

Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

IEEE TRANSACTIONS ON INTELLIGENT TRANSPORTATION SYSTEMS

Framework for Network-Constrained Tracking of Cyclists and Pedestrians

Alphonse Vial[®], Gustaf Hendeby[®], *Senior Member, IEEE*, Winnie Daamen[®], Bart van Arem[®], *Senior Member, IEEE*, and Serge Hoogendoorn[®]

Abstract—The increase in perception capabilities of connected mobile sensor platforms (e.g., self-driving vehicles, drones, and robots) leads to an extensive surge of sensed features at various temporal and spatial scales. Beyond their traditional use for safe operation, available observations could enable to see how and where people move on sidewalks and cycle paths, to eventually obtain a complete microscopic and macroscopic picture of the traffic flows in a larger area. This paper proposes a new method for advanced traffic applications, tracking an unknown and varying number of moving targets (e.g., pedestrians or cyclists) constrained by a road network, using mobile (e.g., vehicles) spatially distributed sensor platforms. The key contribution in this paper is to introduce the concept of network bound targets into the multi-target tracking problem, and hence to derive a network-constrained multi-hypotheses tracker (NC-MHT) to fully utilize the available road information. This is done by introducing a target representation, comprising a traditional target tracking representation and a discrete component placing the target on a given segment in the network. A simulation study shows that the method performs well in comparison to the standard MHT filter in free space. Results particularly highlight network-constraint effects for more efficient target predictions over extended periods of time, and in the simplification of the measurement association process, as compared to not utilizing a network structure. This theoretical work also directs attention to latent privacy concerns for potential applications.

Index Terms—Pedestrians, cyclists, trajectory reconstruction, multiple target tracking, pedestrian tracking, cyclist tracking, road network, road information, moving sensors, data association, multiple hypothesis tracking, network-constrained multihypotheses tracker, NC-MHT, traffic data, traffic monitoring and control.

I. INTRODUCTION

TNFERRING the number of pedestrians and cyclists, as well as their individual states (*e.g.*, position, velocity) from a sequence of measurements, allows drawing a complete

Alphonse Vial, Winnie Daamen, Bart van Arem, and Serge Hoogendoorn are with the Department of Transport and Planning, TU Delft, The Netherlands (e-mail: a.a.vial@tudelft.nl; w.daamen@tudelft.nl; b.vanarem@tudelft.nl; s.p.hoogendoorn@tudelft.nl).

Gustaf Hendeby is with the Division of Automatic Control, Department of Electrical Engineering, Linköping University, 581 83 Linköping, Sweden (e-mail: gustaf.hendeby@liu.se).

Digital Object Identifier 10.1109/TITS.2022.3225467

micro- and macroscopic picture of the traffic flows in an observed area. This is of great value for advanced surveillance applications in traffic operation, control and management.

Traditional setups for these applications rely on stationary or participatory sampling technologies. Stationary sensing systems gather data that can be used to reconstruct trajectories or counts, but only at local scale (*e.g.*, cross- or very short road sections). At the same time, crowd-sourced data from mobile and other wearable devices allow tracking individuals through a network, but require direct or indirect collaboration of the tracked individual and are mostly sparse in nature (not all persons are detected). Current advanced surveillance applications and traffic monitoring typically consider longer time horizons to associate new measurements with existing tracks to monitor individuals over time. Proposed methods mainly consider stationary sensor settings, which often produce a visual stream from a fixed location [1], [2], however come with low spatial resolution and scalability.

With novel distributed-computing and connectivity capabilities, a growing number of on-board sensors, the collection of data on pedestrians and cyclists is expected to further intensify with the advent of connected autonomous mobile systems. Self-driving vehicles, drones, or other types of connected robots will enter populated environments and may act as mobile sensing platforms generating a proliferating amount of data about the platform's internal state, but also about the static and dynamic local area they observe. With the collective intelligence and innate mobility of such sensor platforms, pedestrians and cyclists traffic characteristics could be captured at an extended spatial and temporal scale [3]. In the future, knowledge about position, motion state, and pose of people could enable next generation traffic or crowd surveillance systems to estimate the number of people and reconstruct trajectories across the network.

Yet deriving complete trajectories of indistinguishable pedestrians and cyclists in a network using observations from mobile spatially distributed sensor platforms, is complicated. First, there is an unknown and varying number of people, where position and motion states of individuals are unknown. At the same time, noisy sensors, changing environmental conditions, or occlusion are responsible for missed detections and clutter. And with no *a priori* information about which observations originate from which existing or newly detected individual, the many possibilities of assigning a measurement to an individual complicate the task.

Manuscript received 23 February 2022; revised 13 July 2022 and 6 October 2022; accepted 11 November 2022. This work was supported by the European Research Council and Amsterdam Institute for Advanced Metropolitan Solutions through the ALLEGRO Project under Grant 669792. The work of Gustaf Hendeby was supported by the Center for Industrial Information Technology at Linköping University (CENIIT) under Grant 17.12. The Associate Editor for this article was M. Zhou. (*Corresponding authors: Alphonse Vial; Gustaf Hendeby.*)

There have been major developments in pedestrian and cyclist detection and tracking, for instance, in areas related to autonomous navigation and control. Safety-critical applications, *e.g.*, self-driving vehicles rely on accurate human motion prediction and path planning [4], [5], [6]. Proposed methods and pilot deployments however typically consider spatially restricted short time horizons. Thereby presence and state information is crucial to better understand and anticipate actions, that is knowing what an individual will do next, *e.g.*, start, continue, or stop walking.

In the last decades, different target tracking approaches have been proposed that aim at estimating the number and states of (multiple) dynamic objects using noisy sensor measurements, taking source from the tracking community. Existing algorithms, however, consider short time horizons and are mostly designed for unconstrained motion in two- and three-dimensional space, where complexity tends to grow exponentially due to the number of hypotheses. Attempts have been made to better model the behavior of pedestrians, *e.g.*, [7], which aims at learning to predict for collision avoidance, and [8], in which spatial motion model is learned from observations. Due to their added complexity, these methods fit poorly into standard multi-target tracking methods.

Road information has previously been used to support ground target tracking. The approach proposed in [9] tracks the objects in free space and then, in a wide sense, projects the estimates onto the road network. This ensures that targets are estimated on the road, but does not improve projections as done in this paper. In [10], a variable structure interacting multiple model (VS-IMM) is introduced, which uses different motion models based on the target location to keep the objects on the roads. In [11], targets are described in a combination of global 2-D coordinates, and the quasi 1-D coordinates as used in this paper. This allows for more efficient predictions while, as in [10] and [12], the VS-IMM is used. Compared to the proposed method, the number of target hypotheses are regularly reduced, keeping them alive until observations can determine which one is correct in the proposed multihypotheses tracker (MHT). Furthermore, [10], [11] consider airborne ground moving indicators as sensor, which is different from ground based sensors moving on the road network as introduced in this work. In more recent efforts, [12] uses measurements from static sensors and include road map information, however, focus on tracking interacting vehicles rather than on the effects of road constraints when tracking targets in a network. Reference [13] extends the idea of including road map knowledge to random finite set (RFS) methods, however, using a more involved road representation and handling than the method proposed in this paper. Reference [14] also uses a different road and coordinate representation that requires projections, however, the concept of branching hypotheses in ambiguous situations (e.g., at intersections), also found in [15], is similar to the handling found in our approach.

The key contribution in this paper is to introduce the concept of network bound targets into the multi-target tracking problem. This is done by introducing a target representation, comprising a traditional target tracking representation and a discrete component placing the target on a given edge in the network. This model is then used to derive a networkconstrained multi-hypotheses tracker (NC-MHT). The inclusion of knowledge about the network structure allows for more efficient target predictions over extended periods of time and simplifies the measurement association process, as compared to not utilizing a network structure. The network bound target model is derived with tracking of an unknown and varying number of pedestrians and cyclists in an urban setting using mobile spatially distributed sensor platforms in mind. Hence, the NC-MHT is evaluated on three simulations of targets moving around on an urban infrastructure of connected roads, highlighting different properties introduced by adding the network constraint. The NC-MHT opens up the field for new applications for network-wide traffic surveillance, using information about the number of individuals and their states, to enhance advanced traffic operation, control and management systems.

The remainder of this paper is organized as follows. The next section presents relevant background theory on fundamentals, namely Bayesian filtering and multiple target tracking. Section III introduces the mathematical problem formulation and network-constrained system models. Section IV presents the derived NC-MHT filter and details the integration of the network structure. Section V presents simulation results from three tracking scenarios. Section VI discusses the outlook and potential extensions of the proposed approach. Conclusions are drawn in Section VII.

II. BACKGROUND

Using conventional methods for multiple target tracking (MTT), the individual targets are tracked using Bayesian filters [16], following a step where the available observations in each scan are assigned to the different tracks. In this context, a track represents a potential target and contains information about its estimated past and current state, and a scan is a set of observations received at the specific point in time. A MHT considers several different association hypotheses in parallel. Additional logics handle track creation and termination of tracks over time. The steps are outlined here, and further adapted to network bound targets in the following section.

A. Bayesian Filtering

The state x_k (*e.g.*, position and velocity) of a target at time k can, given the observations from time 1 to k, $z_{1:k}$, be estimated using recursive Bayesian filtering. This is achieved using a two step iterative process that sequentially predicts and updates the probability density function of the target state $p(x_k|z_{1:k})$,

$$p(x_{k+1}|z_{1:k}) = \int p(x_{k+1}|x_k) p(x_k|z_{1:k}) dx_k \quad (1a)$$

$$p(x_k|z_{1:k}) = \frac{p(z_k|x_k)p(x_k|z_{1:k-1})}{p(z_k|z_{1:k-1})}.$$
 (1b)

The first equation predicts future states, given the current state and a motion model $p(x_{k+1}|x_k)$, and the second step

incorporates information from a new observation $p(z_k|x_k)$ into the estimate.

For linear models with Gaussian noise, where both the dynamic and measurement model are linear, the seminal *Kalman filter* (KF) [17] provides the analytic solution to the Bayesian filtering recursion. For the case that nonlinear models are used, the *extended Kalman filter* (EKF) [18], or *unscented Kalman filter* (UKF) [19] can be used to approximate the solution, and in highly nonlinear scenarios the *particle filter* (PF) [20] can be used.

B. Multiple Target Tracking

Multiple target tracking is an extension to the state estimation problem where both the number of targets and their states should be estimated, based on available scans. The observations that make up the scans are noisy, a target may sometimes fail to produce a detection, and clutter (false observations) which all complicate the task.

The core of the MTT problem lies in the association of observations to the right track, as given the association, the state of the tracks can be estimated using Bayesian filtering. Data association is needed as it is not known which observation originates from which target, not to mention which observations are clutter. Classical MTT solutions as the global nearest neighbor (GNN) tracker, the joint probabilistic data association (JPDA) filter, and the multiple hypothesis tracker (MHT) [21] differ in the way the association is performed and how many hypotheses are maintained. See [22] and [23] for more details. More recently developed MTT methods, e.g., the probabilistic hypothesis density (PHD) filter [24], [25], the labeled multi-Bernoulli (LMB) filter [26], and Poisson multi-Bernoulli mixture (PMBM) filter [27], are based on random finite set statistics and are derived differently, resulting in slightly different methods. None of these methods, however, are designed to utilize network constraints.

The MHT is one of the most popular and common MTT methods. The use of several association hypotheses in parallel took off with Reid's seminal paper [21], yet requires many approximations to become a computationally tractable technique as the number of possible track associations increases exponentially with time. Since then, it has been further developed as this technique gained some momentum in the tracking community, and has been applied in specific application domains. The MHT technique relies on evaluating the probabilities of sequences of measurements from various targets. Two main variants of MHT can be found in literature: Hypothesis-Oriented (HO-) [21], and Track-Oriented MHT (TO-MHT) [28]. Cox [29] further contributed to the design of an efficient MHT implementation by introducing Murty's algorithm [30]. The latter reduces the computational complexity as it helps avoiding considering unnecessary hypotheses.

III. PROBLEM FORMULATION

This paper considers a regular MTT problem, with an additional network constraint as imposed, *e.g.*, by targets being bound to a system of roads. That is, given a set of mobile sensors, with limited field of view, determine the number of

targets present in the tracking volume and their respective state. Standard target tracking assumptions are assumed to apply:

- A1: Targets act independently, without influencing one another.
- A2: The number of new targets at each time is Poisson distributed, with the rate λ_{NT} , and targets appear uniformly in the tracking volume.
- A3: In a scan of observations from a sensor, each observation either originates from a single target or is a false detection, and each target produces at most one observation.
- A4: The number of false observations (observations not originating from a target) in a scan is Poisson distributed with the rate λ_{FA} , and the observations are uniformly distributed in the sensors field of view.
- A5: The probability to detect a target that is within the field of a sensor is constant, $P_{\rm D}$.
- A6: The probability a target survives from a time to the next is constant, $P_{\rm S}$.

Additionally, this paper assumes targets to be network constrained, that is:

A7: The targets are bound to network constraints, and hence cannot move freely in the tracking volume. A target is always associated with exactly one discrete network state at any time.

Here, it will be assumed that the network constraint is in fact a road network. However, the theory developed applies to any situation where the targets are associated to a discrete state that can be modeled as a Markov process. Notation will be kept as general as possible.

A. The Network Constraint: Road Network

Targets are constrained to move on roads, sidewalks, and cycle paths, which make up the considered network constraints defined in Assumption A7. A key observation is that a target can only be on one road segment at the time, and once the target reaches an intersection it continues on to another connected road segment, which limits the motion of the targets. This can be modelled as a discrete Markov chain, where the road segments make up the states and the transition probabilities¹ model the probabilities to transition from one road to another once the target reaches an intersection. This can be represented with a directed graph G = (V, A), with a set of vertices V (representing the different states or road segments) of size n and edges $A \subseteq V \times V$ (the possible transition from one road segment to another) of size m. A weight function $p: A \to [0, 1]$ assigns a non-negative probability $p(\delta' | \delta)$ to transition from state δ to segment δ' once the target reaches the end of the road segment represented by δ . It follows from the rule of total probability that $\sum_{\delta'} p(\delta'|\delta) = 1$. Furthermore, each vertex contains information about the length of the road segment it represents, $L(\delta) > 0$. The graph hence represents a topological map of the road network.

¹A discussion on the determination of the transition probabilities is provided in Section VI-B.



Fig. 1. Illustration of the concept showing a simple road network and the underlying Markov Process.

B. Network-Constrained State-Space Model

To utilize the road constraints, a discrete component is added to the target state to indicate which road segment the target is on, and the "regular" part of the state will be used to track the target within the node. Both the prediction and observation model are modified to take the discrete component in consideration.

Targets evolve independently of each other, in discrete time, and the state of each target is represented by a vector

$$X_k = \begin{pmatrix} x_k \\ \delta_k \end{pmatrix},$$

where x_k is a real-valued state vector describing the target motion within a node, and δ_k is the state in the Markov process. The semantics of the state space $x_k \in \mathbb{R}^{n_x}$ can vary depending on the considered application, yet usually includes components related to the target's kinematic state such as position, velocity, and possibly acceleration, along the road segment, but could also be used to describe the lateral position on the road. In the end, the actual position of the target in the world is a combination of the current node δ_k , giving the general position determined from map, and the x_k component which specifies the position along the road segment as illustrated in Fig. 1.

1) *Motion Model:* The nonlinear dynamic model describes the motion within a node as

$$x_{k+1} = f(x_k) + v_k, \quad v_k \sim \mathcal{N}(0, Q_k) \tag{2}$$

where v_k is process noise that is assumed to be white and Gaussian with covariance matrix $\operatorname{cov} v_t = Q_k$. The discrete state δ_k follows the transition probability $p(\delta_{k+1}|\delta_k, x_{k+1})$ where $\delta_{k+1} = \delta_k$ unless x_{k+1} indicates that the target reached the end of the current road segment. Once a state transition occurs, the node specific state, and distance traveled along the current road segment, x_{k+1} is updated accordingly. Typically, this means resetting the distance traveled. In the considered application, when a transition takes place and x_{k+1}^p is the position component of the continuous state, remove the distance traveled on the previous segment to reach the end of it $x_{k+1}^p \leftarrow x_{k+1}^p - L(\delta_k)$. It could, however, potentially also include additional adjustments as a result of specific properties of the new road segment. The adjustment procedure must be repeated if after compensation the target is past the end of the next segment too.

2) Observation Model: At any time instant k, sensor platforms may deliver scans, Z_k . The position of the mobile sensor is assumed to be known, and as defined in Assumption A5, each target is detected with probability P_D when in the field of view of a sensor. The state variables above are related to measurements Z_k according to a nonlinear observation model

$$Z_k = \left(z_k, z_k^{\delta}(\cdot)\right) \tag{3a}$$

$$z_k = h(X_k) + e_k, \quad e_k \sim \mathcal{N}(0, R_k) \tag{3b}$$

$$z_k^{\delta}(\delta') = \Pr(\delta_k = \delta') \tag{3c}$$

where $h(x_k)$ represents the state of the target and relates to an ideal (vector valued) measurement, and e_k is the measurement error which is assumed to be Gaussian with covariance matrix $cov(e_k) = R_k$. Furthermore, each observation includes the probability that the target was observed in the discrete state $\delta', z_k^{\delta}(\delta')$ — a simplification, reasonable in many cases, is that the discrete state information is accurately provided, *i.e.*, that only one state has non-zero probability.

Because several sensors can operate simultaneously, let $S = \{s_v\}_{v=1:V}$ be the set of mobile sensors, where V is the number of operating sensors that acquire data of dynamic targets present on the road network. A scan Z_k^v is the set of all M_k^v measurements received by a sensor platform v at time k such that $Z_k^v = \{Z_k^{v,1}, Z_k^{v,2}, \dots, Z_k^{v,M_k}\}$, and where $Z_k^{v,i}$ represents the *i*th observation received at scan k, by sensor v. This way, a target may be observed by multiple sensor platforms at the same time instant.

C. False Observations and New Targets

As part of the point-to-object and defined in Assumption A3, each target gives rise to at most one measurement per sensor, at each time step. Targets in the field of view are assumed detected and observed with probability $P_{\rm D}$ (Assumption A5). Further, the number of new targets $n_{\rm NT}$ and false observations $n_{\rm FA}$ in the volume V are assumed Poisson distributed,

$$p(n,\lambda,V) = e^{-\lambda V} \frac{(\lambda V)^n}{n!},$$
(4)

with intensity λ_{NT} and λ_{FA} , respectively, see Assumptions A2 and A4. Both new targets and false observations, together denoted extraneous observations, are assumed uniformly distributed in the current tracking volume. The total spatial density of the extraneous observations also follows a Poisson distribution with intensity λ_{EX} and can be expressed as follows

$$\lambda_{\rm EX} \triangleq \lambda_{\rm NT} + \lambda_{\rm FA},\tag{5}$$

which, as will be seen, can be used to simplify some expressions.

Note: In the network constraint cases, the sensors can be assumed to only produce state (road) bound observations, which is beneficial compared to the general case where no such restrictions exist.

VIAL et al.: FRAMEWORK FOR NETWORK-CONSTRAINED TRACKING OF CYCLISTS AND PEDESTRIANS



Fig. 2. Outline of proposed tracking framework, with traditional MHT logic inside orange area.

IV. NETWORK-CONSTRAINED MULTIPLE Hypothesis Tracking

In a MHT, several potential targets are tracked using separate single target tracking filters, and a higher level logic decides which observations to pair with the currently maintained tracks, which observations should be considered to be false, and when to create new tracks. In the MHT, several such hypotheses are maintained in parallel, while at the same time estimating the probability of each of the different hypotheses to be correct. Key components are to generate the appropriate hypotheses, and to reduce the number of considered hypotheses as much as possible to keep the computational complexity as low as possible. The network constraint, being discrete in its nature, simplifies the hypothesis handling. This section outlines the MHT formulas adopted to take the network constraints into consideration. Fig. 2 provides an overview of the different elements of the proposed algorithm.

A. Extended Kalman Filter

The EKF [18] is arguably the most common technique used for single target tracking in the MHT. The EKF does not deal with discrete components, hence it cannot be directly applied to the hybrid state proposed in Sec. III-B.

1) *Time Update:* The solution is to condition on the discrete state δ_k , and use the regular EKF for the continuous part of the state x_k . This yields, for a given δ_k , for the time update

$$\hat{x}_{k|k-1} = f(\hat{x}_{k-1|k-1}),$$
(6a)

$$P_{k|k-1} = F_k P_{k-1|k-1} F_k^T + Q_k.$$
(6b)

where $\hat{x}_{k|k-1}$ is the mean of the estimate of the state at time k given the measurements up until time k-1, $P_{k|k-1}$ is the matching error covariance matrix, and $F_k = f'(\hat{x}_{k-1|k-1})$ the linearization around $\hat{x}_{k-1|k-1}$ of the dynamic model.

Caveat, performing this prediction step might position the target past the end of the current discrete state. The interpretation in the described setting is that the target reaches the end of the current road segment. That is, the target should transition to a new discrete state. This is done following the underlying Markov model as described in Section III-A. If there are several different possible transitions, where each have transition probabilities, all are taken and new track hypotheses are created, and the hypothesis probability is updated accordingly by the MHT logics, as described below. After the transition the continuous state x is compensated in each transition, as described in Sec. III-B. As for the case there is no discrete state to be propagated to, *e.g.*, a target is existing the tracking volume of interest, the target is simply removed from the set.

2) Measurement Update: When an observation is obtained, again conditioned on the discrete state, the continuous part of the state can be updated, provided the discrete state is possible in the observation, using

$$\hat{x}_{k|k} = \hat{X}_{k|k-1} + K_k(z_k - \hat{z}_k),$$
(7a)

$$P_{k|k} = P_{k|k-1} - K_k H_k P_{k|k-1}, \tag{7b}$$

$$\hat{Z}_k = h(\hat{x}_{k|k-1}) \tag{7c}$$

$$S_k = H_k P_{k|k-1} H_k^T + R_k, \tag{7d}$$

$$K_k = P_{k|k-1} H_k^T S_k^{-1}, (7e)$$

where $H_K = h'(\hat{x}_{k|k-1})$. If no observation is available, the update step is simply skipped, and if several are available the step is repeated.

Contrary to the prediction step, the observation update will never trigger transition in the discrete state, if the simplified version is used. The update is performed under the presumption that the observation places the target in the current discrete state, hence the estimate stays in the same state. The only ambiguities in this step is due to different observations being associated with the current track.

3) Corner Cases: The above approach assumes that each track hypothesis can fully be assigned to a discrete state. This is often a reasonable approximation as long as the estimated track is far from the end of the current road segment, as compared to the uncertainty of the estimate. However, if the track is close to the edge of the road segment or the uncertainty is considerable, with a non-negligible probability the track could instead be on an adjacent road segment. Two different solutions to this are considered here: always splitting tracks

with significant overlap, and splitting when needed for proper association with observations.

In the former case, when a track is predicted close to an intersection, split the track in n - 1, where n is the number of potential roads at this intersection, and divide the probability according to the probability mass on all segments other than the segment the target is currently on (based on the uncertainty). The resulting two new tracks should have means combining to the original mean, and a combined covariance (including the spread of the mean term) should match the original covariance. A benefit of this approach is that it is easy to implement, however, it leads to unnecessarily fast growth of hypotheses to consider.

In the latter case, observations are allowed to be associated with tracks on adjacent discrete states, if there is a significant probability of leakage. Associated this way, the track is split into two identical ones splitting the probability as above, and where the one is projected back at the extension of the adjacent segment (*i.e.*, as if a transition had taken place but x_k had not yet been adjusted) and then that track is updated as if it had been on the segment. This results in fewer hypotheses, as splitting is only performed when needed, but the logic is slightly more complex to implement.

4) Track Representation: As should be clear from the presentation above, a specific estimated track, *i*, is fully described by the sequence of discrete state transitions assumed to take place, $\delta_{1:k}^{i}$, and the observations associated with the track, $\theta_{1:l}^{i}$, together denoted $\tau_{k|l}^{i} = (\delta_{1:k}^{i}, \theta_{1:l}^{i}), k \ge l$ for the state at time *k* with measurements from up until time *l*. And several different $\tau_{k|l}$ could describe the same underlying target, with different assumptions about the discrete state sequence and what observations originate from the target.

New tracks are started where available measurements are not associated with any other existing track. The observation is used to set the general position determined from the map and the position on the road segment such that given Z_k ,

$$x_k = h^{-1}(z_k) \tag{8a}$$

$$\delta_k = z_k^o(\delta') = \Pr(\delta_k = \delta'). \tag{8b}$$

If the state x_k cannot be uniquely determined from z_k (*h* is not invertible), as much as possible is derived from z_k and remaining parts of the state vector is set to default values. As an example, assume x_k state with position and velocity and measurements z_k providing (possibly via a conversion) the position; then the position part of x_k is given by z_k and the speed is set to a nominal value, *e.g.* 0. This represents the possibility that the observation corresponds to a new target that has entered or newly appeared in the network.

B. Track Hypotheses and Scores

A track comprises the different hypotheses about a target, as a result of different $\tau_{k|l}$ assumptions as described above, and a score for each of these. The different track hypotheses assumed to represent the same underlying target can be represented with a track tree, where the tree branches are a result of discrete state transitions and how the observations are associated. The score indicates the importance of the specific hypothesis. For track hypothesis i this is described as the log-likelihood ratio [31]

$$\ell_{k|l}^{i} = \log \frac{p(\text{target exist}|\tau_{k|l}^{i}, Z_{1:l})}{p(\text{no target}|\tau_{k|l}^{i}, Z_{1:l})}.$$
(9)

The score can be updated recursively, as decision points are reached in the track tree.

Similar to the track state, the track score update can be divided into two steps. When the track is propagated in time, except for compensating for the probability to survive affecting all tracks alike, only the changes of discrete states affect the score when the score is divided amongst the potential transitions. The score of track hypothesis *i* becomes

$$\ell_{k|k-1}^{i} = \ell_{k-1|k-1}^{p(i)} + \log(P_{\rm S}p(\delta_k|\delta_{k-1}, \hat{x}_{k|k-1}^{i})), \quad (10)$$

where p(i) is the parent hypothesis (*i.e.*, a single prior hypotheses) of hypothesis *i*. It should be noted that, when no state transition is possible $\log(p(\delta_k|\delta_{k-1}, x_{k|k-1}^i)) = \log(P_s)$. If $P_s = 1$ this means the score remains unchanged.

The update of the score as the result of assigning a new observation (or not) as the result of obtaining a scan is a bit more involved,

$$\ell_{k|k}^{i} = \ell_{k|k-1}^{p(i)} + \begin{cases} \log(1 - P_D), & \text{if } \theta_k^{i} = \emptyset\\ \log\left(\frac{P_D p(Z_k|Z_{1:k-1}, \tau_{k|k}^{i})}{\lambda_{\text{FA}}}\right), & \text{if } \theta_k^{i} \neq \emptyset \end{cases},$$
(11)

where

$$p(Z_k|Z_{1:k-1}, \tau_{k|k}^i) = p(z_k|Z_{1:k-1}, \tau_{k|k}^i) \cdot z_k^{\delta}(\delta_k)$$

= $\mathcal{N}(z_k|\hat{z}_k^i, S_k^i) \cdot z_k^{\delta}(\delta_k).$ (12)

There are two different cases: an observation is obtained or not.

If no observation is obtained, $\theta_k^i = \emptyset$, the only change to the score is that the probability of the target is scaled by the probability to not observe the target, $(1 - P_D)$. That is, the score drops if the target is expected to have been observed.

If an observation is obtained, $\theta_k^i \neq \emptyset$, the probability of the target is compensated by the probability to observe the target, P_D , and the probability that the target would produce the obtained observation (which can be computed from the innovation in the Kalman filter). The no target probability is scaled with the probability of obtaining the observation without a target present, λ_{FA} .

Note: The absence of observations, sometimes denoted negative information, represents a valuable source of information as it lowers the score for track hypotheses where the target was not observed as expected. Hence, though the track state does not change, it is important to update the track scores based on this negative information. This helps to reduce the number of track hypotheses that should be considered.

New tracks are created from observations and given the initial score $\ell_{k|k}^i = \log(\gamma_{\text{NT}})$, where γ_{NT} is used to get the



Fig. 3. Illustration of track hypotheses and global hypotheses. A target tree can be constructed to store the hypothetical tracks of the corresponding target. A global hypothesis represents a set of all possible tracks that are compatible from different targets. The state transition occurs when x_{k+1} indicates that the target reached the end road segment, here indicated by the state δ^0 .

right initial likelihood ratio. Terminating tracks, on the other hand, are simply dropped by cause of being too unlikely.

Given the track score $\ell_{k|l}^i$, the probability of track *i* existing can be computed as

$$p(\text{target } i \text{ exists}|Z_{1:l}, \tau_{k|l}^{i}) = \frac{e^{\ell_{k|l}^{i}}}{1 + e^{\ell_{k|l}^{i}}}, \quad (13)$$

where it has been utilized that the probability of existence and non-existence must sum to 1.

C. Global Hypotheses

A global tracking hypothesis, $T_{1:k}$, is a description of a complete solution to the multi-target tracking problem; i.e., the number of targets and their respective states. A global hypothesis assigns each observation to a track, or assumes it is a false observation, and determines the sequence of discrete states of all tracks. This can also be described as picking out which track hypotheses to use. The number of track hypotheses grows rapidly over time, as ideally all possible discrete state transitions and measurement associations should be explored. However, only a few of these track hypotheses are consistent with each other, fulfilling Assumption A3 which states that a measurement can only originate from one target, and can form global hypotheses. A global hypothesis, therefore, contains at most one hypothetical track from each track tree. Fig. 3 illustrates track trees, track hypotheses and global hypotheses, and how they are updated as new scans of observations arrive.

The probability of a global hypothesis is obtained as the product of the probability of each included track hypothesis and the probability of all unassociated measurements being false alarms. The log-probability of the global hypothesis l can be computed as the sum of the score of the contained track

hypothesis compensated with a term taking false observations into consideration

$$\log\left(p(\mathbf{T}_{1:k}^{l}|Z_{1:k})\right) = \sum_{i \in \operatorname{tracks}(l)} \ell_{k|k}^{i} + |Z_{1:k}|\log(\lambda_{\mathrm{FA}}), \quad (14)$$

where tracks(*l*) is the set of track hypotheses in global hypothesis *l*, and $|Z_{1:k}|$ the total number of observations.

The global track score can be updated recursively, using the recursive property of the track hypotheses scores

$$\log(p(\mathbf{T}_{1:k}^{l}|Z_{1:k})) = \log(p(\mathbf{T}_{1:k-1}^{l}|Z_{1:k-1})) + \sum_{i \in \text{tracks}(l)} \ell_{k}^{i} + |Z_{k}|\log(\lambda_{\text{FA}}), \quad (15)$$

where $\ell_k^i = (\ell_{k|k}^i - \ell_{k-1|k-1}^{p(i)})$ is the score increment in the current step for track hypothesis *i*.

In this formulation, the network constraints are naturally handled, as they are contained in the track scores, and do not require any special handling.

D. Complexity Saving Methods

To achieve a computationally tractable solution, it is necessary to lower the number of considered global hypotheses. This is done by removing too improbable hypotheses, and if possible only generate relevant hypotheses. The methods used for this are discussed in this section.

1) Generating the k-Best Hypotheses: The number of global hypotheses grows exponentially with time, and needs to be handled to achieve computational tractability. One way to limit the complexity is to only generate a subset of the possible hypotheses. That is, to construct a subset of data associations $\tilde{\Theta}_k \in \Theta_k$ possible in each step, such that $|\tilde{\Theta}_k| \ll |\Theta_k|$, which will generate the best scores. This can be posed as an assignment problem and solved with standard



Fig. 4. Road-constrained gating.

combinatorial optimization algorithms. Many algorithms exist to efficiently solve the optimal assignment problem based on the additive increments in the track scores, *e.g.*, the Hungarian algorithm [32], the Auction algorithm [33], and the Jonker-Volgenant-Castanon [34] algorithm. If combined with Murty's algorithm [30], the *k*-best assignments can be found in polynomial time. The remaining hypotheses can then simply be left unconsidered.

2) *Gating:* Gating is a technique to simplify the association problem, making it less computationally demanding by removing potential associations that are too unlikely. This is achieved by ignoring track to observation associations, if the observation lies outside the gate of the track.

The gate is set such that the probability of rejecting a true association is low. Using ellipsoidal gating [35] the gate is defined by

$$D_{\rm G} = (z_k - \hat{z}_{k|k-1})^T S_k^{-1} (z_k - \hat{z}_{k|k-1}) \le \gamma_{\rm G}, \qquad (16)$$

which defines an ellipsoid around the predicted observation in which the observation must be considered. Assuming Gaussian distributions, $D_{\rm G} \sim \chi^2(n_z)$, it is possible to design $\gamma_{\rm G}$ for a given probability to reject a correct association.

With the hybrid state space, as considered here, gating is performed in two steps, first observations where the discrete state does not match are dropped, and then normal gating is performed. Fig. 4 exemplifies the gating procedure, which is simplified by the fact that the continuous part of the observation is scalar such that

$$|z_k - \hat{z}_{k|k-1}| \le \kappa \sigma_{k|k-1},\tag{17}$$

where κ is a factor times standard deviations, $\sigma_{k|k-1}$ in the considered dimension.

3) Clustering: Clustering is a method in which tracks that share observations are put in clusters, and then each cluster is treated independently. This is beneficial as association and track hypotheses generation scales poorly with the number of involved targets and observations. When producing the final global hypotheses, hypotheses from the different clusters can be combined freely. The clustering is performed based on which tracks allow the same observation in their gate.

Clusters can be formed recursively over time. In that case, two clusters that gate in the same observation must be merged. Similarly, clusters should regularly be examined to see if they can be split, as a result of removing track hypotheses that have connected sub-clusters. For more details and implementation details the reader is referred to [21], [28], and [36]. Once again, the hybrid nature of the suggested model can help simplify the handling.

4) Pruning Strategies: Track, Hypothesis and Target Management: Tracks are deleted based on the track score [37], or N-scan sliding window [38]. The first method prunes low probability hypotheses and tracks. N-scan pruning, on the other hand, traces back to the node at scan k - N and deletes the subtrees that diverge from the selected branch at that node. This represents the possibility that a target has disappeared from the scene. Furthermore, as indicated in Section IV-A, the network constraints cause tracks that can not be propagated to a new discrete state to be removed from the set of existing tracks. This represents the possibility that a target has exited the network. Targets are also removed accordingly, that is as soon as it has no remaining track in any hypothesis. Because the track score contain all relevant statistical information, remaining tracks lose no information during the pruning process.

V. SIMULATION STUDY

In this section, the results from an empirical evaluation on three simulated scenarios are presented. The scenarios are chosen to highlight properties of the proposed NC-MHT on track-handling at intersections, data association, and measurement dependency. To evaluate the proposed approach, the NC-MHT filter was integrated into the standard (unconstrained) MHT filter from [36], to which it is compared² in terms of tracking performance. Note that it is difficult to provide any accurate complexity analysis, and that the proposed method has low maintenance to keep branching tracks alive, that is, carries more hypotheses that can be easily handled. The comparison is performed in terms of summarizing statistics of the trackers and the MTT measure *generalized optimal sub-pattern assignment* (GOSPA) metric [39],

$$d_p^{(c,2)}(\mathbf{x}, \hat{\mathbf{x}}) = \left(\min_{\gamma \in \Gamma_{(i,j) \in \gamma}} d(xi, \hat{x}_j)^p + \frac{c^p}{2} (|\mathbf{x}| + |\hat{\mathbf{x}}| - 2|\gamma|) \right)^{\frac{1}{p}}$$
(18)

where $\hat{\mathbf{x}} = {\hat{x}_1, \dots, \hat{x}_{|\hat{\mathbf{x}}|}}$ is a finite subset of state estimates of the ground truth set $\mathbf{x} = {x_1, \dots, x_{|\mathbf{x}|}}$, and γ represents the assignment sets between these two sets. Let $d^c(x, \hat{x}) =$ min($||x - \hat{x}||, c$) denote a cutoff metric for any $x, \hat{x} \in \mathbb{R}^n$. Further, let Γ denote the set of all possible assignment sets. GOSPA allows a decomposition of the error into three parts: 1) localisation error, 2) missed targets, and 3) false targets. In the simulations, the GOSPA is computed based on the 2D positions of the tracks, and the parameters are set to c = 8, corresponding to a maximum error appropriate for this setting, and p = 2, corresponding to using the 2-norm which is a standard choice.

Note that GOSPA is a single unified performance metric, whereas the other quantities are presented because they represent properties that are important in MTT.

 $^{^{2}}$ The compared trackers are implemented in Python and C++, and the simulations were run on a laptop with a 2.8 GHz Quad-Core Intel Core i7 and 16 GB memory.



(a) Illustration of NC-MHT and the standard two-dimensional MHT, where ground truth is represented as solid green line. The inclusion of network constraints allows for more efficient target prediction over longer time periods, and simplifies the measurement association.



(b) Tracking results at different time steps, visualizing the position and uncertainty of estimated tracks in two dimensions. Note that 2-D allows to visualize the evolution of the [NC-MHT] uncertainty, while it kept only 1-D (*i.e.*, along the road) in the implementation. The blue crosses are sensor measurements and are at the center of the sensor's field of view (blue circle).

Fig. 5. Illustration and simulation results of main differences in tracking behaviour between NC-MHT (black crosses) and the standard MHT (red diamonds).

A. Simulation Setup

The kinematic state at time k is described by the target's position and velocity. For simplicity one-dimensional motion is considered for the NC-MHT such that $x_k = [y_k, \dot{y}_k]^T$ contains position and velocity. Nonlinear measurement models are more adequate when using real-world data, as not only position and velocity of a target is measured, but also range (r), bearing (α) , and angle rate (\dot{r}) to a target's position can

be measured (*e.g.*, radar). The choice of an appropriate model depends on the considered application, yet this implementation uses linear-Gaussian dynamics and observations as handling the nonlinear case is a trivial extension following (6)–(7). The kinematic state motion model follows linear constant velocity (CV) given parameters

$$F = \begin{bmatrix} 1 & \Delta_T \\ 0 & 1 \end{bmatrix}, \quad Q = q^2 \cdot \begin{bmatrix} \Delta_T^3/3 & \Delta_T^2/2 \\ \Delta_T^2/2 & \Delta_T \end{bmatrix}$$

with sampling time 1 s, and where Δ_T 0.1 m is the process noise variance. In the = q simulation, targets move with initial mean speed 1.415 m/s (standard deviation is 0.215 m/s). Nonlinear effects in the target dynamics at intersections are ignored in this simulation, but can easily be considered using models describing interactions between, e.g., targets at higher density junctions. For simplicity the probability of target survival is set to $P_{\rm S} = 1$. The discrete state transition probabilities, moving targets from segment to segment, are equally distributed among the set of possible outgoing transitions from any given node.

In the simulation, sensors are mobile and independent with initial mean speed 12.3 m/s (standard deviation is 1.5 m/s), and limited field of view [-30, 30] (m) in longitudinal direction. The designed path for targets (*e.g.*, a sidewalk) is assumed all observed within a sensor's field of view. The parameters of the measurement model are

$$H = \begin{bmatrix} 1 & 0 \\ 0 & 1 \end{bmatrix}, \quad R = \begin{bmatrix} \sigma^2 & 0 \\ 0 & \sigma^2/4 \end{bmatrix}$$

where $\sigma = 0.5$ m. The clutter intensity is $\lambda_{FA}(z) = \lambda V U(z)$ where $U(\cdot)$ is a uniform density over the sensor field of view, $V = 2 \times 30$ m is the network-constrained "volume" of the field of view, and $\lambda = 0.01 \times m^{-1}$ is the average number of clutter observations per unit volume. Further, the discrete state information included in each provided observation is accurate. In situations where several sensor platforms operate simultaneously, a sequential measurement update strategy is employed. That is, a separate measurement update step is applied for each sensor. This approach scales well with the number of sensors, and fits naturally in a distributed setting as measurements can be dealt with as they arrive.

The NC-MHT performance is compared to a standard MHT that describes the kinematic state by the target's position and velocity in two dimensions, and where the motion follows a constant velocity model. Similarly, sensor observations are given in Cartesian x- and y-coordinates.

Note: For the simulation, target and clutter observations were generated in the network-constrained space, yet used for both NC-MHT and standard MHT when comparing both performances. This ensures using the same measurements when comparing both filters, however, it considerably benefits the two-dimensional standard MHT. Also, the choice of linear models benefits the performance of the standard MHT, as compared to the NC-MHT.

B. Variable Negative Information

To save computations this implementation exploits the vital knowledge provided by so-called negative information,



Fig. 6. True target tracks (in different colors) for the three simulated scenarios, where black lines are road segments. In scenario 1 (left), one target is initiated on the horizontal segment before the fork junction, which eventually leads to a track split. In scenario 2 (center), two targets are initiated well separated, before entering the two horizontal segments, potentially passing close to each other. In scenario 3 (right), three targets are initiated on random segments and more targets are born from different locations during the simulation.

as described in Section IV-B. Regularly providing the filter with measurements of targets that should have been seen by the sensor, *i.e.*, non-presence, has significant impact on track likelihood scores and can help abandon unlikely tracks. Hence, the MTT should be updated with all scans, empty or not to maintain proper track scores. It is tempting to drop empty scan, as they do not affect the track estimates just the scores, in order to save time. To evaluate the importance of negative information and see its effect in the presented setting, a scheme with a varying proportion of empty scans reported to the MTT is proposed in the third scenario.

C. Results

Three simulated scenarios were used in this study. The first two are designed to highlight the benefits of the proposed NC-MHT as compared to the standard MHT when it comes to target prediction and measurement association. The third scenario evaluates the measurement dependency of the filter given a different number of observations. For the first two scenarios 500 Monte Carlo runs are performed, whereas the third, including two variations, is run 100 times each. The presented results are averaged over the Monte Carlo runs. For all three simulations, the maximum number of hypotheses per cluster and the minimum normalized hypothesis score are set to 50 and 4 respectively. Also note that only tracks that have been assigned at least twice with a measurement have been kept for presentation in the global hypothesis.

Scenario 1 In the first scenario, one target initiated ahead of the fork junction was simulated for 100 time steps. True trajectories are shown in Fig. 6. For this scenario, 10 sensor platforms were initiated randomly distributed in the network. The scenario parameters were set to $P_{\rm D} = 0.95$, $\lambda_{\rm NT} = 0$, $\lambda_{\rm FA} = 0.6$.

This scenario particularly highlights beneficial network constraint effects for more efficient target prediction over an extended period of time. This scenario is challenging because when the target leaves the fork junction, and no measurement update is performed, correctly predicting the target state is complicated. The network constraints allow to keep track of the two different turns the target may perform as it leaves the fork intersection, whereas the standard filter predicts the target to continue straight and going off-road eventually losing track, unless an observation is obtained right after the turn. This behaviour is illustrated in Fig. 5a, where the standard MHT misses the turn in absence of new updates and continues its prediction in the wrong direction ($\tau_1^1 - \tau_5^1$). With the arrival of new measurements, the MHT reacts by initiating a new track with along the new road segment ($\tau_6^3 - \tau_7^3$).

The GOSPA performance is shown in Fig. 9, and summarizing statistics are shown in Fig. 7. A closer look at the latter unveils a higher number of tracks produced by the NC-MHT, here resulting in a higher number of clusters and global hypotheses. This a likely bi-product of the clutter model benefiting the standard MHT and can be improved by adjusting tuning parameters. Overall, for this scenario the GOSPA results show that the NC-MHT gives smaller errors than the standard MHT filter. The GOSPA location error is larger for the NC-MCT towards the end, this is a result of the NC-MHT being better at maintaining tracks, whereas the MHT lose them effectively moving part of the location error to a missed track error. This is further substantiated by Table II. Hence, in cases where maintaining track continuity is of importance the proposed NC-MHT has benefits.

Scenario 2 In the second scenario, two targets are initiated well separated before moving onto a horizontal segment in opposite direction were simulated for 100 time steps. True trajectories are shown in Fig. 6. For this scenario, 20 sensor platforms were initiated randomly distributed in the network. The scenario parameters were set to $P_{\rm D} = 0.95$, $\lambda_{\rm NT} = 0$, $\lambda_{\rm FA} = 0.6$.

This setup highlights how adding network constraints simplifies the measurement association process. This scenario is challenging for traditional approaches because when the targets are close their measurements form a single cluster, making data association difficult. Given the type of targets, their speeds are assumed low and thus do not allow to distinguish their movements in two different directions that way. Fig. 5 illustrates how traditional approaches without network restriction need to consider all incoming measurements (if in the gate) for the measurement to track association, and

VIAL et al.: FRAMEWORK FOR NETWORK-CONSTRAINED TRACKING OF CYCLISTS AND PEDESTRIANS



Fig. 7. Statistics for simulation scenario 1.



Fig. 8. Statistics for simulation scenario 2.

possibly fail. The network constraints allow to bypass growing free-space complexity of traditional approaches and lead to more efficient gating and data association as measurements as motion is bound to a specific segment.

The GOSPA performance is shown in Fig. 10, and summarizing statistics are shown in Fig. 8. Overall, for this scenario the GOSPA results show that the NC-MHT performs better than the standard filter. The NC-MHT can keep tracks longer in time, as shown in Table II. Note that, as compared to the first scenario, Scenario 2 has twice the number of sensors, hence resulting in more observations. This could explain the lower location error. As expected, results suggest that NC-MHT number of misses significantly less targets.

Scenario 3 In the third scenario, three targets initiated at random locations in the network are simulated for 100 time steps. New targets enter in the surveillance area at different time steps and are born at random locations in the network. Targets leaving the surveillance area is not considered here, however, can be easily handled by adjusting the probability of



Fig. 9. Tracking results for simulation scenario 1.



Fig. 10. Tracking results for simulation scenario 2.

survival P_S for future implementations. The true trajectories are shown in Fig. 6. For this scenario, the performance of the filters were compared based on two realisations with different numbers of: a) number of available sensors, and b) proportion of empty scans provided to the tracker. The scenario parameters are shown in Table I.

This scenario illustrates a slightly more realistic urban traffic setting and is challenging because of the many intersections, targets, and observations. This simulated scenario particularly highlights the potential of the approach for real-world applications. More precisely, Scenario 3a) highlights the (low) density of observations needed as a result of efficient predictions along the road segments in the proposed method. Scenario 3b), on the other hand, explores the importance of the negative information provided to the tracker.

Fig. 11 shows the GOSPA performance of the NC-MHT filter given 5, 10, 20 and 40 mobile spatially distributed sensor platforms respectively, and the performance of the MHT with 40 initiated sensors. Results show that the NC-MHT achieves a

IEEE TRANSACTIONS ON INTELLIGENT TRANSPORTATION SYSTEMS



 TABLE I

 PARAMETERS FOR THE TWO REALISATIONS OF SIMULATION SCENARIO 3

Fig. 11. Tracking results for simulation Scenario 3a). The legend refers the number of sensors initiated.

similar performance to the standard filter yet with significantly less sensors. In this simulation, the lower the number of sensors, the fewer measurements are provided to the tracker, and the less updates can be performed by the filter. This confirms results from scenario 1 suggesting that the included network structure knowledge of the NC-MHT allow for more efficient target predictions over extend period of time. The difference in track length shown in Table II supports this outcome. More sensors means more clutter measurements, but also more observations of true targets. Results also show that with more sensors, more unlikely tracks can be killed, *e.g.*, existing tracks get less likely when sensors expected a target but did not see any.

Fig. 12 shows the GOSPA performance of the NC-MHT filter given 25, 50, 75, and 100 percent of available sensors that provide the tracker with additional negative information, respectively, and is compared to the performance of the MHT given all sensors are included in the time update step. The results indicates that adding only a few instances of negative information can benefit the tracker significantly.

VI. OUTLOOK

In this section, we discuss technical and non-technical effects that may be induced by design decisions on the tracker, and the tracking system.

A. Motion and Target Dynamics

A comprehensive understanding of targets motion and their respective environment is an important factor for better trajectory reconstruction. Extending the proposed framework using more sophisticated kinematic models can easily be achieved,



Fig. 12. Tracking results for simulation scenario 3b). The legend refers to available proportion of available sensors (in percent) that provide the tracker with additional negative information, respectively.

and could lead to better prediction performances on the shortstandard models are widely accepted for their term. As simplicity, it seems important to keep their accuracy for trajectory estimation in balance with inherent computational cost. Yet, the longer the prediction horizon needs to be, the more important choice of the model becomes. Moreover, it is about how to make better predictions when little information is available. On the other hand, improving the scene understanding using social- and behaviour-aware models could significantly strengthen tracking and resulting trajectory estimations (e.g., [25]), for instance, in shared spaces, when approaching signalized intersections, or during overtaking maneuvers. This additional complexity however is a non-trivial extension to our proposed MHT approach as it breaks the assumption of independent targets. However with some approximations, e.g., that the interactions do not interfere with anything else than the propagation of targets, these models could probably be used with good results and at a reasonable cost. With pedestrian and cyclist motion being influenced by weather, time of day, surrounding infrastructure and environment, as well as social and behavioural cues, future work would benefit from including such contextual information as this can lead to better longterm, and potentially network-wide, tracking results. Future extensions of this work could also substitute fixed parameters with learned distributions, which helps the tracker to better interpret measurements and tracks (e.g., [8]). For instance, the rate at which new targets appear and where they appear are dependent on time and space and could similarly be learned.

B. Sensing Infrastructures and Measurements

The mobile sensing setting used in this work can conceptually be extended to any type of sensing realm where incoming measurements provide information about targets, and regularly report non-presence to improve tracking. Also since current developments hint to the integration of additional decentralized processing layers where computing power is brought to the edge, thereby fusion of additional information from additional VIAL et al.: FRAMEWORK FOR NETWORK-CONSTRAINED TRACKING OF CYCLISTS AND PEDESTRIANS

RESULTS FOR ALL SCENARIOS 1–3										
	GOSPA		NLE		МТ		FT		Track length	
	NC-MHT	MHT	NC-MHT	MHT	NC-MHT	MHT	NC-MHT	MHT	NC-MHT	MHT
S1	411.7	848.3	224.5	199.3	217.5	2838.8	2396.9	4732.0	52.8	26.9
S2	159.2	860.0	37.5	1086.2	52.5	4532.6	708.7	2264.0	10.3	5.7
S3.a - 5	1095.5	n.a.	1313.1	n.a.	9192.3	n.a.	2477.4	n.a.	27.1	n.a.
- 10	1004.5	n.a.	1611.6	n.a.	6389.7	n.a.	3072.9	n.a.	33.0	n.a.
- 20	857.3	n.a.	1404.3	n.a.	3927.0	n.a.	3127.3	n.a.	36.4	n.a.
- 40	693.7	1482.5	894.6	198.8	1890.8	9758.7	3182.0	13027.5	39.2	37.1
S3.b - 25	1055.1	n.a.	1467.5	n.a.	3787.2	n.a.	7058.2	n.a.	58.1	n.a.
- 50	944.1	n.a.	1545.1	n.a.	3693.1	n.a.	4827.8	n.a.	49.0	n.a.
- 75	897.5	n.a.	1556.5	n.a.	3836.8	n.a.	3778.8	n.a.	42.2	n.a.
- 100	878.5	1485.4	1425.3	185.9	3931.8	11152.6	3421.7	11883.2	37.8	32.0

TABLE II RESULTS FOR ALL SCENARIOS 1-

static or participatory sensor infrastructure is conceivable and represents no extra adaptation to the proposed method.

Contextual information about the static and dynamic environment can be captured by different types of sensors. Using additional information could lead to significant improvements of the tracking system, and particularly in the data association logic, for instance as a way to strengthen the discrimination between different objects. Recent advances in deep-learningbased trackers, for instance, improve the discriminative power for visual tracking applications [40].

Furthermore, node transition probabilities can either be determined based on prior knowledge or set based on empiric studies about, *e.g.*, how many take a left or right turn at an intersection. Following traffic patterns this could be something that changes throughout the day and week and could be function of environment features (*e.g.*, weather, green canopy), real-time incidents (*e.g.*, events, construction), network knowledge (*e.g.*, road types, touristic routes) or infrastructure (*e.g.*, road quality).

C. Privacy Considerations

Further improvements of the NC-MHT tracking performance can make it more suited for real-world applications, yet the gain in utility may come at the cost of the targets' privacy. The centralised scheme presented in this work, which pools all location observations to a central entity and hence may facilitate the surveillance of individuals or collectives, raises privacy concerns.

Individual location traces are akin fingerprints, as demonstrated by past research on the similarity of human mobility patterns [41], their predictability [42], and the uniqueness of individual traces [43]. But it is the realisation that with sufficient data it is possible to precisely locate, track, and possibly (re-)identify people that is a major source of concern. Standard MTT approaches have been frequently used in the location privacy community, *e.g.*, Gruteser and Hoh were among the first to use a MHT to link anonymized location samples to some individual users and recreate their traces [44].

Anonymized location samples as considered in this paper, however, create a false sense of security because of the the spatial and temporal correlation between successive observations. Longer tracking duration, for instance, typically leads to an accumulation of information and can help infer sensitive information about individuals. The traffic density, on the other hand, can have both positive and negative impact on privacy, but is a factor beyond an individual's control. Using auxiliary data sources (*e.g.*, household address or location-based service databases) to the anonymous traces further increases the risk of inferring locations, identities, and potentially interests of an individual.

Overall, the accumulation of observations in a central database, anonymized or not, reduces the burden for attacks that can easily be automated and applied to large numbers of individuals, therefore augmenting the risk beyond individual tracking to a more collective threat.

More work is needed to study effects of the methodological, and higher-level system design space, both on target privacy and utility of future applications.

VII. CONCLUSION

The advent of mobile sensor platforms is expected to drastically increase the amount of information collected about traffic participants (*e.g.*, pedestrians or cyclists) in populated environments, which can be of great value for multiple future applications, e.g., digital twin, smart road infrastructure, autonomous driving, and advanced traffic surveillance. To make use of this information, this work presented a framework for network-constrained tracking of targets with observations from spatially distributed and mobile sensor platforms. The key contribution represents the introduction of network bound targets into the multi-target tracking problem.

The proposed approach can be applied to any type of network-constraint environment, bypassing the growing free-space complexity of traditional MTT approaches, and use different types of observations. The generic nature of this work makes it interesting for any type of sensing setting where incoming observations provide information about targets, and potentially report non-presence of targets. Derived information can, for instance, be valuable input for traffic signal controller aiming at reducing idling times for cyclists at signalized intersections, for crowd monitoring systems measuring previously unobserved network links, or fusing observations from different sensors and road users to distribute traffic at the micro level. Overall, we see the potential to utilize this information in more places as it can be applied to more difficult problems than those addressed before.

Future work will explore the application space for next generation traffic surveillance and control systems, evaluating a range of simulated scenarios using a microscopic traffic simulator. The highlighted properties of the modified method

IEEE TRANSACTIONS ON INTELLIGENT TRANSPORTATION SYSTEMS

as compared to a standard MHT can be used to predict the gains obtainable with other MTT filters. A more thorough evaluation, then focused on a particular scenario or application is an interesting direction for future research. Furthermore we motivate the need for experiments in real road traffic scenarios, which can help increase the persuasiveness of the proposed framework. It should be noted that given the (at this time) difficult access to data from a fleet of moving sensor platforms over longer time and space, any other dataset that provides information about targets and regularly report non-presence could be applied to our framework. This work hints to future directions for research in tracking and privacy engineering within the intelligent transportation realm.

REFERENCES

- J. Barthélemy, N. Verstaevel, H. Forehead, and P. Perez, "Edgecomputing video analytics for real-time traffic monitoring in a smart city," *Sensors*, vol. 19, no. 9, p. 2048, May 2019.
- [2] D. C. Duives, T. Van Oijen, and S. P. Hoogendoorn, "Enhancing crowd monitoring system functionality through data fusion: Estimating flow rate from Wi-Fi traces and automated counting system data," *Sensors*, vol. 20, no. 21, p. 6032, Oct. 2020.
- [3] A. Vial, W. Daamen, A. Y. Ding, B. Van Arem, and S. Hoogendoorn, "AMSense: How mobile sensing platforms capture pedestrian/cyclist spatiotemporal properties in cities," *IEEE Intell. Transp. Syst. Mag.*, vol. 14, no. 1, pp. 29–43, Jan. 2022.
- [4] A. Rudenko, L. Palmieri, M. Herman, K. M. Kitani, D. M. Gavrila, and K. O. Arras, "Human motion trajectory prediction: A survey," *Int. J. Robot. Res.*, vol. 39, no. 8, pp. 895–935, 2020.
- [5] C. G. Keller and D. M. Gavrila, "Will the pedestrian cross? A study on pedestrian path prediction," *IEEE Trans. Intell. Transp. Syst.*, vol. 15, no. 2, pp. 494–506, Apr. 2014.
- [6] J. Kooij, F. Flohr, E. Pool, and D. Gavrila, "Context-based path prediction for targets with switching dynamics," *Int. J. Comput. Vis.*, vol. 127, no. 3, pp. 239–262, Mar. 2019.
- [7] I. Batkovic, M. Zanon, N. Lubbe, and P. Falcone, "A computationally efficient model for pedestrian motion prediction," in *Proc. Eur. Control Conf. (ECC)*, Jun. 2018, pp. 374–379.
- [8] M. Luber, G. D. Tipaldi, and K. O. Arras, "Place-dependent people tracking," Int. J. Robot. Res., vol. 30, no. 3, pp. 280–293, 2011.
- [9] B. Pannetier, K. Benameur, V. Nimier, and M. Rombaut, "Ground moving target tracking with road constraint," *Proc. SPIE*, vol. 5429, pp. 138–149, Apr. 2004.
- [10] T. Kirubarajan, Y. Bar-Shalom, K. R. Pattipati, and I. Kadar, "Ground target tracking with variable structure IMM estimator," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 36, no. 1, pp. 26–46, Jan. 2000.
- [11] M. Ulmke and W. Koch, "Road-map assisted ground moving target tracking," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 42, no. 4, pp. 1264–1274, Oct. 2006.
- [12] D. Song, R. Tharmarasa, T. Kirubarajan, and X. N. Fernando, "Multivehicle tracking with road maps and car-following models," *IEEE Trans. Intell. Transp. Syst.*, vol. 19, no. 5, pp. 1375–1386, May 2018.
- [13] J. Zheng and M. Gao, "Tracking ground targets with a road constraint using a GMPHD filter," *Sensors*, vol. 18, no. 8, p. 2723, Aug. 2018.
- [14] J. López-Araquistain, Á. J. Jarama, J. A. Besada, G. D. Miguel, and J. R. Casar, "A new approach to map-assisted Bayesian tracking filtering," *Inf. Fusion*, vol. 45, pp. 79–95, Jan. 2019.
- [15] K. Krishanth, R. Tharmarasa, T. Kirubarajan, P. Valin, and E. Meger, "Prediction and retrodiction algorithms for path-constrained targets," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 50, no. 4, pp. 2746–2761, Oct. 2014.
- [16] H. Fang, N. Tian, Y. Wang, M. Zhou, and M. A. Haile, "Nonlinear Bayesian estimation: From Kalman filtering to a broader horizon," *IEEE/CAA J. Autom. Sinica*, vol. 5, no. 2, pp. 401–417, Mar. 2018.
- [17] R. E. Kalman, "A new approach to linear filtering and prediction problems," J. Basic Eng., vol. 82, pp. 35–45, Oct. 1960.
- [18] A. H. Jazwinski, Stochastic Processes and Filtering Theory (Mathematics in Science and Engineering), vol. 64. Amsterdam, The Netherlands: Elsevier, 1970.
- [19] S. J. Julier and J. K. Uhlmann, "New extension of the Kalman filter to nonlinear systems," *Proc. SPIE*, vol. 3068, pp. 182–193, Apr. 1997.

- [20] B. Ristic, S. Arulampalam, and N. Gordon, *Beyond the Kalman Filter: Particle Filters for Tracking Applications*. Norwood, MA, USA: Artech House, 2004.
- [21] D. B. Reid, "An algorithm for tracking multiple targets," *IEEE Trans. Autom. Control*, vol. AC-24, no. 6, pp. 843–854, Dec. 1979.
- [22] S. S. Blackman and R. Popoli, *Design and Analysis of Modern Tracking Systems*. Norwood, MA, USA: Artech House, 1999.
- [23] Y. Bar-Shalom and X. R. Li, *Estimation and Tracking: Principles*, *Techniques, and Software*. Norwood, MA, USA: Artech House, 1993.
- [24] B. N. Vo and W. K. Ma, "The Gaussian mixture probability hypothesis density filter," *IEEE Trans. Signal Process.*, vol. 54, no. 11, pp. 4091–4104, Nov. 2006.
- [25] K. Krishanth, X. Chen, R. Tharmarasa, T. Kirubarajan, and M. McDonald, "The social force PHD filter for tracking pedestrians," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 53, no. 4, pp. 2045–2059, Aug. 2017.
- [26] S. Reuter, B. T. Vo, B. N. Vo, and K. Dietmayer, "The labeled multi-Bernoulli filter," *IEEE Trans. Signal Process.*, vol. 62, no. 12, pp. 3246–3260, Dec. 2014.
- [27] Å. F. García-Fernández, J. L. Williams, K. Granström, and L. Svensson, "Poisson multi-Bernoulli mixture filter: Direct derivation and implementation," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 54, no. 4, pp. 1883–1901, Aug. 2018.
- [28] T. Kurien, "Issues in the design of practical multitarget tracking algorithms," in *Multitarget-Multisensor Tracking: Advanced Applications*, Y. Bar-Shalom, Ed. Norwood, MA, USA: Artech House, 1990, ch. 3, pp. 43–83.
- [29] I. J. Cox and S. L. Hingorani, "An efficient implementation of Reid's multiple hypothesis tracking algorithm and its evaluation for the purpose of visual tracking," *IEEE Trans. Pattern Anal. Mach. Intell.*, vol. 18, no. 2, pp. 138–150, Feb. 1996.
- [30] K. G. Murty, "An algorithm for ranking all the assignments in order of increasing cost," *Oper. Res.*, vol. 16, pp. 682–687, Mar. 1968.
- [31] Y. Bar-Shalom, S. S. Blackman, and R. J. Fitzgerald, "Dimensionless score function for multiple hypothesis tracking," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 43, no. 1, pp. 392–400, Jan. 2007.
- [32] H. W. Kuhn, "The Hungarian method for the assignment problem," Naval Res. Logist., vol. 2, nos. 1–2, pp. 83–97, Mar. 1995.
- [33] D. P. Bertsekas, "The auction algorithm: A distributed relaxation method for the assignment problem," Ann. Oper. Res., vol. 14, no. 1, pp. 105–123, Dec. 1988.
- [34] R. Jonker and A. Volgenant, "A shortest augmenting path algorithm for dense and sparse linear assignment problems," *Computing*, vol. 38, no. 4, pp. 325–340, Nov. 1987.
- [35] J. B. Collins and J. K. Uhlmann, "Efficient gating in data association with multivariate Gaussian distributed states," *IEEE Trans. Aerosp. Electron. Syst.*, vol. 28, no. 3, pp. 909–916, Jul. 1992.
- [36] J. Olofsson, E. Brekke, T. I. Fossen, and T. A. Johansen, "Spatially indexed clustering for scalable tracking of remotely sensed drift ice," in *Proc. IEEE Aerosp. Conf.*, Mar. 2017, pp. 1–13.
- [37] S. S. Blackman, "Multiple hypothesis tracking for multiple target tracking," *IEEE Aerosp. Electron. Syst. Mag.*, vol. 19, no. 1, pp. 5–18, Jan. 2004.
- [38] C. Kim, F. Li, A. Ciptadi, and J. M. Rehg, "Multiple hypothesis tracking revisited," in *Proc. IEEE Int. Conf. Comput. Vis. (ICCV)*, Dec. 2015, pp. 4696–4704.
- [39] A. S. Rahmathullah, A. F. Garcia-Fernandez, and L. Svensson, "Generalized optimal sub-pattern assignment metric," in *Proc. 20th Int. Conf. Inf. Fusion (Fusion)*, Jul. 2017, pp. 1–8.
- [40] L. Zhang, H. Han, M. Zhou, Y. Al-Turki, and A. Abusorrah, "An improved discriminative model prediction approach to real-time tracking of objects with camera as sensors," *IEEE Sensors J.*, vol. 21, no. 15, pp. 17308–17317, Aug. 2021.
- [41] L. Alessandretti, U. Aslak, and S. Lehmann, "The scales of human mobility," *Nature*, vol. 587, no. 7834, pp. 402–407, Nov. 2020.
- [42] C. Song, Z. Qu, N. Blumm, and A.-L. Barabási, "Limits of predictability in human mobility," *Science*, vol. 327, no. 5968, pp. 1018–1021, 2010.
- [43] Y.-A. De Montjoye, C. A. Hidalgo, M. Verleysen, and V. D. Blondel, "Unique in the crowd: The privacy bounds of human mobility," *Sci. Rep.*, vol. 3, p. 1376, Mar. 2013.
- [44] M. Gruteser and B. Hoh, "On the anonymity of periodic location samples," in *Security in Pervasive Computing* (Lecture Notes in Computer Science), D. Hutter and M. Ullmann, Eds. Berlin, Germany: Springer, 2005, pp. 179–192.



Alphonse Vial is currently pursuing the Ph.D. degree with the Department of Transport and Planning, TU Delft. His work investigates the potential of emerging data capturing systems for traffic research and practice, while aims to evaluate the impact on individuals and society. His current research interests lie in the broad areas of intelligent transportation systems, mobile cyber-physical systems, sensor fusion, urban data and its applications, and privacy engineering.



Bart van Arem (Senior Member, IEEE) received the M.Sc. and Ph.D. degrees in applied mathematics from the University of Twente, The Netherlands, in 1986 and 1990, respectively. From 1991 to 2009, he worked at TNO. From 2003 to 2012, he was a Part-Time Full Professor at the University of Twente. He was appointed as a Full Professor Transport Modeling at the Department of Transport and Planning, Faculty of Civil Engineering and Geosciences, TU Delft, in 2009. He was the Head of the Department Transport and Planning from 2010 to 2017 and

served as the Director of the TU Delft Transport Institute from 2012 to 2021. He serves as a Pro Vice Rector Magnificus for Doctoral Affairs of TU Delft since 2021. He is the Founding Editor-in-Chief of the IEEE OPEN JOURNAL OF INTELLIGENT TRANSPORTATION SYSTEMS.



Gustaf Hendeby (Senior Member, IEEE) received the M.Sc. degree in applied physics and electrical engineering and the Ph.D. degree in automatic control from Linköping University, Linköping, Sweden, in 2002 and 2008, respectively. He worked as a Senior Researcher at the German Research Center for Artificial Intelligence (DFKI) (2009–2011), as a Senior Scientist at the Swedish Defense Research Agency (FOI) and held an adjunct associate professor position at Linköping University (2011–2015). He is currently an Associate Professor and a Docent

with the Division of Automatic Control, Department of Electrical Engineering, Linköping University. His main research interests are sensor fusion and stochastic signal processing with applications to nonlinear problems, target tracking, and simultaneous localization and mapping (SLAM). He is the author of several published articles and conference papers in the area. He has experience of both theoretical analysis as well as implementation aspects. Since 2021, he has been a Leader of the WASP Localization and Navigation Area Cluster. In 2022, he served as the General Chair for the 25th IEEE International Conference on Information Fusion (FUSION), Linköping. Since 2018, he has been an Associate Editor for IEEE TRANSACTIONS ON AEROSPACE AND ELECTRONIC SYSTEMS in the area of target tracking and multisensor systems.



Winnie Daamen is currently an Associate Professor with the Department of Transport and Planning, Delft University of Technology, The Netherlands. Her research interests include theory, modeling, and simulation of traffic (pedestrians, cyclists, vehicles, and vessels). The core of her expertise is in data collection, data analysis to get insight into traffic participant behavior, and modeling these behaviors. This does not only involve pedestrians, but also cyclists, drivers, and autonomous vehicles.



Serge Hoogendoorn was appointed Antonie van Leeuwenhoek Professor Traffic Operations and Management in 2006. He has been the Chair of the Department of Transport and Planning since 2018. He is currently (one of the four) a Distinguished Professor of Smart Urban Mobility with the Delft University of Technology. He has a part time appointment at Monash University. He is an Honorary Professor with the School of Transportation, Southeast University, China. He is the PI Mobility with the Institute of Advanced Metropolitan Solu-

tions (www.ams-amsterdam.com). His current research evolves around smart urban mobility, with focal areas such as i) theory, modeling, and simulation of traffic and transportation networks, including cars, pedestrian, cyclists, and novel public transport services; ii) development of methods for integrated management of these networks; iii) impact of uncertainty of travel behavior and network operations; iv) impact of ICT on network flow operations, robustness, and resilience, and v) urban data and its applications. In all these topics, his work considers both recurrent and non-recurrent situations.